Author's response to reviews

Title: The Selective Cytotoxic Anti Cancer Properties and Proteomic Analysis of Trigonella Foenum-Graecum

Authors:

Abdulaziz Alsemari (alsemari@kfshrc.edu.sa)
Fahad Alkhodairy (alhkodairy@kfshrc.edu.sa)
Ahmad Aldakan (Aldakan@kfshrc.edu.sa)
Mai Al-Mohanna (Al-Mohanna@kfshrc.edu.sa)
Eman Bahoush (Bahoush@kfshrc.edu.sa)
Zakia Shinwari (Shinwari@kfshrc.edu.sa)
Ayodele Alaiya (AAlaiya@kfshrc.edu.sa)

Version: 7 Date: 22 December 2013

Author's response to reviews: see over
22/12/2013

Dear

Mr James Prozenko
on behalf of Dr William CS Cho
Executive Editor
BMC Complementary and Alternative Medicine

We are submitting the revised manuscript and the reviewer’s comments response. In response to the last reviewer’s comments, the manuscript was edited by an English speaking editor and all the language corrections were applied. Thanks

Regards

Abdulaziz Alsemari

*Abdulaziz AlSemari, MD
Department of Neurosciences
King Faisal Specialist Hospital and Research Centre.
Riyadh 11211 PO box 3354. Saudi Arabia

Tel 0096614424804. Fax 0096614424763
alsemari@kfshrc.edu.sa